Author's response to reviews

Title: Diffuse duodenal nodular lymphoid hyperplasia: a large cohort of patients etiologically related to Helicobacter pylori infection

Authors:

Mehnaaz S Khuroo (mkhuroo@yahoo.com)
Naira S Khuroo (naira_sultan@yahoo.com)
Mohammad S Khuroo (khuroo@yahoo.com)

Version: 8 Date: 5 April 2011

Author's response to reviews:

I have done necessary major revisions as desired.
All references 1 to 48 are shown in the text as square brackets as per journal guidelines.
Some references were missing in the text which have been included. At present all 48 references listed are listed in the text sequentially.
Typhographical errors have been rechecked and corrected.
I hope this completes the necessary changes and the manuscript is now put online as it has been accepted nearly a month back and any further delays shall be unnecessary.
If you need any clarifications you can send me the return mail and I shall respond immediately.